Author's response to reviews

Title: Awareness of Human papillomavirus and factors associated with intention to obtain HPV vaccination among Korean youth: Quasi experimental study

Authors:

haewon Kim (haewon@snu.ac.kr)

Version: 3 Date: 22 May 2014

Author's response to reviews: see over
Dear Editor,

I am hereby submitting the manuscript entitled “Awareness of Human papillomavirus and factors associated with intention to obtain HPV vaccination among Korean youth: Quasi experimental study” for consideration for publication as an Article in the BMC International Health & Human Rights. I believe that the findings of this study are relevant to the scope of this journal and will be of interest to its readership. The study examined the awareness among fifth-grade girls and boys of sexually transmitted diseases (STDs), cancer, and human papillomavirus (HPV), and to determine the factors associated with intention to obtain the HPV vaccination.

This paper is not being considered for publication in any other journal. Furthermore, the manuscript contains only original data that have not been published elsewhere either in part or in their entirety. As the author of this manuscript, I have read the paper, attest to the validity of its contents, and agree to its submission to the journal. And there are no conflicts of interest to declare.

This research was supported by Basic Science Research Program through the National Research Foundation of Korea (NRF) funded by the Ministry of Science, ICT & Future Planning(grant number: 2012- R1A1A301-5525). Thank you for considering this manuscript for publication in the BMC International Health & Human Rights and I look forward to hearing from you.

Yours sincerely,
Corresponding author:
Hae Won KIM (RN, PhD)
Associate Professor
The research institute of nursing science, College of Nursing, Seoul National University,
Taehakro 103, Jongro-Gu, Seoul,
South Korea 110-799
Tel.: +82-2-740-8820; Fax: +82-2-765-4103
Email: haewon@snu.ac.kr, haewon63@gmail.com